Author's response to reviews

Title: BACE1 inhibitory activity of fungal endophytic extracts from Malaysian medicinal plants

Authors:

Azzeme Harun (azzaly83@yahoo.com)
Richard Muhammad Johari James (richard@salam.uitm.edu.my)
Siong Meng Lim (steven79@yahoo.com)
Abu Bakar Abdul Majeed (abubakar@salam.uitm.edu.my)
Anthony LJ Cole (tony.cole@canterbury.ac.nz)
Kalavathy Ramasamy (kalav922@salam.uitm.edu.my)

Version: 4 Date: 6 June 2011

Author's response to reviews: see over
06 June 2011

Ms Iratxe Puebla
BioMed Central Editorial Team

Dear Ms Iratxe Puebla,

Re: BACE1 inhibitory activity of fungal endophytic extracts from Malaysian medicinal plants

Thank you for the email concerning the above manuscript. We have included the identification of all four fungal isolates as requested. We have amended the manuscript and have highlighted the corrections in yellow.

We hope the amendments made are to the satisfaction of the Editorial Board and the paper is accepted for publication.

Thank you.

Yours sincerely,

[Signature]

Kalavathy Ramasamy
Faculty of Pharmacy,
Universiti Teknologi MARA
Selangor Darul Ehsan
Malaysia